

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Economidou, Claire; Karamanis, Dimitris; Kechrinioti, Alexandra; Xesfingi, Sofia

Article

The role of social capital in shaping Europeans' immigration sentiments

IZA Journal of Development and Migration

Provided in Cooperation with:

IZA – Institute of Labor Economics

Suggested Citation: Economidou, Claire; Karamanis, Dimitris; Kechrinioti, Alexandra; Xesfingi, Sofia (2020) : The role of social capital in shaping Europeans' immigration sentiments, IZA Journal of Development and Migration, ISSN 2520-1786, Sciendo, Warsaw, Vol. 11, Iss. 1, pp. 1-31, https://doi.org/10.2478/izajodm-2020-0003

This Version is available at: https://hdl.handle.net/10419/298675

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Claire Economidou^{1,*}, Dimitris Karamanis², Alexandra Kechrinioti³, Sofia Xesfinqi⁴

The Role of Social Capital in Shaping Europeans' Immigration Sentiments

Abstract

Migration has manifested itself to historic highs, creating divisive views among politicians, policy makers, and individuals. The present paper studies the Europeans' attitudes toward immigration, focusing particularly on the role of social capital. Based on 267,282 respondents from 22 countries and over the period 2002–2014, we find that despite the eventful past years, Europeans, on average, are positive toward immigrants with the North European countries to be the least xenophobic. A salient finding of our analysis is that regardless of the impact of other contextual factors, namely, a country's macroeconomic conditions, ethnic diversity, cultural origin, and individuals' attributes, social capital associates with positive attitudes toward all immigrants, independent of their background. Furthermore, social capital moderates the negative effects of perceived threat on people's opinions about immigrants.

Current version:	November 07, 2018
Keywords:	immigration, social capital, public attitude, survey, Europe
JEL codes:	C25, F22, J61, O52
Corresponding author:	Claire Economidou
	economidou@unipi.gr

¹ Department of Economics, University of Piraeus, Piraeus 185 34, Greece. E-mail: economidou@unipi.gr

² Department of Economics, University of Piraeus, Piraeus 185 34, Greece. E-mail: karamanis@unipi.gr

³ Department of Economics, University of Piraeus, Piraeus 185 34, Greece. E-mail: kechrinioti@unipi.gr

⁴ Department of Economics, University of Piraeus, Piraeus 185 34, Greece. E-mail: xesfingi@unipi.gr

[©] The Author(s). 2020. Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. Cite as: Economidou et al. *IZA Journal of Development and Migration* (2020) 11:3.

1 Introduction

Migration has strongly manifested itself to historic highs.¹ Spurred by civil war, political instability, and the global financial crisis, constantly growing immigration flows to European and between European countries posed certain challenges for host countries and rose concerns about the potential costs of welcoming more immigrants on employment, welfare benefits, security, and social cohesion.

Two concurring factors have arguably contributed to bringing migration to the center of public attention. First, the aftermath of dramatic economic recession left European countries fragile, with weaker economic prospects, generalized cuts to public spending, and voters to question the legitimacy of foreign nationals' presence in their countries. Second, the Syrian exodus and the number of people who have arrived in Europe in the last few years to seek asylum having fled conflicts and oppression in Asia and Africa has reached historic highs. The absence of supranational EU-level coordinated mechanisms to respond to the emergency led many member countries to autonomous and uncoordinated responses, with national political concerns and hostile feelings to be on the rise among the general public.

Immigration has also been highly politicized. It took, for instance, center stage in the campaign that led to the vote for the "Brexit" of the UK from the European Union in June 2016. It has also been debated in many other European countries (Austria, Hungary, the Netherlands). In addition, terrorist attacks in France, Belgium, and the UK perpetuated by second-generation immigrants triggered a harsh debate that questioned the idea of integration of foreign nationals in hosting societies. Immigration, and in particular undocumented migration, has been a divisive issue in the US political debates.

Yet, while the economic recession seemed to have provided a justification for political pandering to a surge of anti-immigrant sentiment, it is far from clear how much public opinion has really shifted in this direction. This article studies what shapes Europeans' sentiments toward immigration. We carefully combine rich individual survey and aggregate data to analyze the factors and conditions that contribute to cross-country attitude variation toward immigration and explore the relevance of theories in explaining the formation of these attitudes.

Our work relates to an insightful body of research on public sentiments on immigration that has developed in recent years since the seminal cross-national work of Quillian (1995).² Subsequent research has studied the role of individuals' skills (Mayda, 2006); cultural elements; threats to national identity; racism (Schlueter and Wagner, 2008; Rustenbach, 2010; Markaki and Longhi, 2013); compositional amenities, such as characteristics of neighbors and co-workers (Card et al., 2012); religion (McDaniel et al., 2011); values and beliefs (Hainmueller and Hiscox, 2007), ethnic diversity (Brader et al., 2008); language difference (Hopkins, 2011); conservative social attitudes (Ford, 2011); security fears (Lahav and Courtemanche, 2012); crime concerns (Fitzgerald et al., 2012); state of governance, that is, direct or representative democracy (Hainmueller and Hangartner, 2015); redistributional effects of welfare state (Facchini and

According to the Migration Policy Institute, the total annual asylum applications in the European Union (EU) Member States and European Free Trade Association countries from 2008 to 2015 have increased by 444% – from 256,155 to 1,393,285 applications. Only in 2015, there were 3.8 millions of new immigrants in the EU, half of them originating from non-member countries.

² The seminal work of Quillian (1995) examined the impact of macroeconomic circumstances on attitudes to immigrants in Europe and regressed a measure on prejudice against immigrants on the inverse of gross domestic product (GDP) per capita across 12 European countries.

Mayda, 2009); and personal traits and predisposition (Dinesen et al., 2016) in shaping immigration sentiments and voting preferences (Dustmann et al., 2019).³

The purpose of this paper is to study the role of trust, networks, norms, and sanctions facilitating collective action for mutual benefit – the so-called social capital – along with relevant contextual factors on immigration attitudes. Social capital is considered to be the "glue" for cooperative and equalitarian societies, which are characterized by better citizens, i.e., citizens who are more engaged in community life, trust each other and their institutions, are tolerant and respectful of others and their rights (Putnam, 2000), have better governance (Bowles and Gintis, 2002), and have higher economic growth prospects (Knack and Keefer, 1997). Therefore, in our increasingly multicultural societies if sound social capital generates more integrated societies, it should be relevant for immigration.

Although social capital is an expanding research theme in economics, there has been some suspicion across the entire theoretical spectrum, from neoclassical (Arrow, 1999; Durlauf, 1999) to Marxist (Fine, 2010).⁴ We argue that one reason of concern for social capital is the mismatch between theoretical coverage of the concept and empirical work that severely has undermined the usefulness of social capital as a tool for economic research. The reason is that social capital is a multidimensional concept (Paxton, 1999; Putnam, 2000; Sabatini, 2008, 2009), and in most empirical applications, the definition is largely data driven and is limited by the very narrow range of proxies that the chosen data set contains.⁵ Collapsing, however, social capital to narrow proxies may be inadequate and leads to erroneous conclusions.

We therefore contribute to the literature in three district ways. First, we demonstrate empirically the multifacet nature of social capital by including various dimensions of it to explore all potential channels it works. In this respect, our work differs from other relevant important studies on social capital and immigration that mainly explore a single dimension of social capital. For example, Herreros and Criado (2009) and Rustenbach (2010) studied one aspect of social capital, that of interpersonal trust, whereas Côté and Erickson (2009) explored the role of associations and networks on immigration.

Second, to better explore the role of social capital, we enrich the individual-level data with macro-level data. The inclusion of economic and multicultural profile of a country allows one to obtain more insights about the mechanisms at work and discuss competing theories. For example, weak macroeconomic condition (recession, financial vulnerabilities, inequality) and multicultural societies may trigger a rather dark side of social capital, the so-called "block-ing capital" – the use of community cohesion to reinforce the exclusion of outsiders through the mechanisms of social capital (Portes and Landolt, 1996). For example, the presence of out-group, such as that of the immigrants, may pose a threat to the in-group (Blumer, 1958).

³ The study of Dustmann et al., 2019 uses a quasi-random allocation design and data from Danish regions to derive causal impacts of refugee allocation on voting. The authors found that refugee allocation has a considerable effect on voting outcomes and there is a clear divide in the political responses of urban and rural populations to refugee allocation.

⁴ Perhaps, the suspicion by some economists about the concept of social capital stems from the fact that it has emerged from sociological traditions. For example, Durlauf and Fafchamps (2004) addressed the issue of "conceptual vagueness"; Arrow (1999) had suggested that the term be abandoned, largely due to his rejection that it constitutes a form of "capital"; and Fine (2002) criticized social capital as a ahistorical and apolitical concept.

⁵ Alesina and La Ferrera (2000), for example, used a single dimension that of being membership in voluntary organizations from the US General Social Survey (GSS). Kan (2007) relied only on a measure of whether or not people think there is someone living nearby that would help them in an emergency from the US Panel Study of Income Dynamics, and Rocco and Fumagelli (2014) used the generalised trust.

Therefore, we aim to test whether social capital pertains its bright side and has a positive direct relation with immigration attitudes and further whether it has a positive indirect relation, that is the mitigation of typical "perceived threat" conditions (such as recession, unemployment) in a society.

Finally, it is likely to have a two-way causality between social capital and attitudes toward immigration, as well as attitudes may also have a direct influence on social capital. Further, there may be unobserved factors that influence both of these variables; social capital is usually measured at the individual level by choice variables (such as participation, affiliation, and trust) that will be influenced by unobserved preferences. In our empirical work, we focus on the association between Europeans' sentiments on immigration and social capital, but we also explore the robustness of our results using the (two-stage residual inclusion) control function approach (Wooldridge, 2015).⁶

Our empirical analysis covers 22 European countries, uses survey data from the European Social Survey (ESS) – a large-scale biennial study of attitudes and values – for a total of 267,282 individuals over the period 2002–2014 and develops around two main questions: (i) What explains Europeans' xenophile(/phobic) attitudes toward immigration? and (ii) do Europeans' attitudes vary across different immigrant profiles?

Despite the economic upheavals, we find Europeans, on average, to be rather positive toward immigrants with the North European countries to exhibit the least xenophobic attitude. A finding that emerges strong in our analysis is that regardless of the impact of other contextual factors, countries with high social capital exhibit more positive attitudes toward all immigrants. We therefore find a direct, strong positive association between social capital and immigration stance. Furthermore, social capital also exhibits an indirect association with immigration attitudes, as it appears to moderate the negative effects of adverse economic situations on attitudes toward immigrants. Our findings corroborate with the important role the literature has assigned to social capital in increasing welfare and growth (Knack and Keefer, 1997). In increasingly multicultural societies, social capital has an additional contribution: it makes societies more receptive to immigration.

Such findings could be important for the increasingly multicultural societies we live in. Rather than cultivating differentiation, fragmentation, and exclusion, successful public policies should aim at investing in social capital. Policies that foster the bonding of different communities as well as bonding between communities and public agents increase country's social cohesion and prosperity.

At the outset of this paper, we would like to stress some potential limitations. As we mentioned earlier, like the vast majority of literature in this area, endogeneity poses a challenge and one should be cautious in referring to causal relationships between social capital and immigration sentiments; the issue of endogeneity has been carefully outlined elsewhere (see, e.g., Côté and Erickson, 2009). However, in this study, we make an effort to alleviate such concerns by following appropriate econometric techniques and explore various instruments that other relevant studies do not (see, e.g., Côté and Erickson, 2009; Herreros and Criado, 2009; Rustenbach, 2010). Another constraint that potentially limits our conclusions is the nature of the ESS

⁶ Gannon and Roberts (2018) applied a two-stage residual inclusion estimation technique to study the association between social capital and health. The study of Dustmann et al., 2019 is another one, among the very few, that deals with endogeneity and uses a quasi-random allocation design (instead of IV) to derive causal impacts of refugee allocation on voting.

database, which provides cross-sectional information and does not allow one to explore more variation in the data or apply dynamic techniques. Lack of information on skill composition of immigrants relative to natives did not allow us to delve deeper into different skill profiles of the immigrants and explore how sentiments change based on skills. Last but not the least, an important limitation of our analysis is that we lack information on member diversification of networks and associations, which would be important to identify potential dark (i.e., non- community-enhancing) sides of social capital. Acknowledging that, we are careful in studying the average effect of social capital, as some of its aspects (e.g, types of associations) may contribute to less tolerance against minority groups, but we are unable to unmask it in the present study.

The remainder of the article proceeds as follows. Section 2 introduces the framework for modeling individuals' attitudes to immigration and the estimation technique. Section 3 presents the data. Section 4 discusses the results. Section 5 summarizes the findings and concludes.

2 A framework of analysis

2.1 Social capital and immigration attitudes

Social capital as a theoretical concept is present within both sociological and economic traditions.⁷ Conceptualizing social capital is a very difficult task. This is because, as Coleman (1988, 1990) – a leading sociologist in his study of social capital on the creation of human capital – pointed out, social capital exists between actors; it is not attributable to an actor like human and economic capital are. In general, sociological work has focused more on understanding social capital in a conceptual sense than on measuring it. However, there are exceptions to this and Putnam is a significant example, presenting a huge amount of empirical evidence for the US and popularizing the concept of social capital. Putnam (1993, pp. 35–36) defined social capital as "features of social organization, such as trust, norms, and networks, that can improve the efficiency of society by facilitating coordinated actions". To measure social capital, Putnam (2000) looked at associational behaviors such as voting participation, church attendance, recreational activities, and membership in civic groups, including, most famously, bowling leagues.

The ideas of shared trust, norms, and values have also a long tradition in economics. Bruni and Sugden (2000) pointed out that in his Lectures on Jurisprudence, Adam Smith presents a theory of social capital that is quite similar to that of the modern economic sociological theories of Putnam (1993, 2000).⁸ Arrow (1972) showed how social connections can compensate for expensive formal structures in facilitating financial transactions, and Kreps et al. (1982) showed how increased interaction facilitates cooperation. Most of the theoretical emphasis of the economic social capital literature has been on trust and how trust can improve the efficiency of social exchange (Bellamare and Kroger, 2004; Bowles and Gintis, 2002).

In similar vein, the scant literature on immigration sentiments and social capital focuses mainly on one aspect of the latter, that of trust. For example, the study of Herreros and Criado

⁷ One of the first formal uses of social capital as an academic concept was laid out by the French Marxist sociologist Bourdieu (1986), which refers to economic resources that one gains from being part of a network of social relationships, including group membership.

⁸ Smith argued that "... reputations for trustworthiness are transmitted through networks of trading relationships; the denser the network ... the greater is the value of reputation and so ... the greater is the degree of the trust." (Bruni and Sugden, 2000, p. 33)

(2009) employed one dimension of social capital, that of "trust in people", to study the role of social trust on attitudes toward immigration. The study argues that societies with high levels of social capital facilitate the integration of immigrants because those members with high levels of social trust will tend to have more positive attitudes toward immigration. This hypothesis is empirically tested in a sample of 16 European countries, drawing on the 2002–2003 wave of the ESS. Results show that, regardless of the impact of other factors such as levels of unemployment and percentage of foreign population, citizens with high social capital do exhibit more positive attitudes toward immigration than the rest of the population.

The role of social integration and its effects on anti-immigrant attitudes were also studied by Rustenbach (2010) among other explanations, namely, cultural marginality theory, human capital theory, political affiliation, neighborhood safety, contact theory, foreign investment, and economic competition. Results reveal that among all tested explanations, interpersonal trust was the strongest predictor of anti-immigrant attitudes. Individuals with high levels of interpersonal trust are more likely to overcome the uncertainty associated with the unknown. In addition, findings show that living in regions where interpersonal trust is high may create a positive context of perception toward immigrants and influence an individual toward a more positive view.

A prominent aspect of the social capital theory is the aggregate value of citizen participation in associations and organizations, social ties and networks, civic engagement, trust, and norms of reciprocity. After all, the whole point of social capital is to facilitate collective action. It is usually bandied as a cure for commons-type problems, such as littered parks and crimeinfested streets, but there is no reason to assume that a part of society will deploy its social capital only for wise and benevolent ends. A neighborhood, for instance, that is adept at organizing litter patrols and crime watches will also likely be adept at organizing opposition to real estate developments that threaten to add economic or racial diversity. In this respect, the forces that bind communities together may also have a dark side, the so-called "blocking capital" (Portes and Landolt, 1996), i.e., the use of community cohesion to reinforce the exclusion of outsiders. In other words, social capital can act as a double-edged sword; social capital within a group is often built by excluding those outside that group.

The study of Côté and Erickson (2009) explored this aspect of social capital in studying the relationship between social networks, associations, and tolerance. By constructing an index of network and association diversity, the authors aim to unmask the potential negative aspects of social capital. Findings from the analysis of the 2000 Canadian federal election show that different forms of network diversity and voluntary associations have very different relationships with tolerance. Diversity of ties to middle-class people is found to be related to greater tolerance and is consistent with the higher education levels of middle-class people and their lower degree of economic competition with minorities, both of which contribute to higher tolerance. Diversity of ties to working-class people is associated with lower tolerance and is consistent with their lower education levels and greater economic competition with minorities. Voluntary associations exhibit more complex picture. Associations with well-educated members or a relatively high proportion of minority members positively relate to tolerance because they include kinds of people disposed to tolerance in contrast to associations with poorly educated memberships. However, often the associations as such have no effect at all. Our study adds to the literature by considering more dimensions of social capital than the aforementioned studies. Along with other contextual factors, our paper explores the role of social capital in shaping immigration sentiments, bringing evidence for more than a decade and for a large panel of countries.

More particularly, we consider three wide dimensions of social capital, as proposed by Coleman (1988): (i) trust in people and institutions; (ii) interpersonal networks of informal type, such as meeting with friends, relatives, and colleagues; participating in associations, including churches; and voluntary work (so-called Putnam-type social activities) or formal type, such as work for political parties or professional organizations among others (so-called Olson-type social activities); and (iii) norms and effective sanctions. We include 13 aspects across the three dimensions of social capital to capture the multifacet nature of social capital. These aspects are trust in people, trust in institutions (trust legal system, state of health services, state of education system), networks (social interaction, church attendance, worked in political/action group, contacted politician/government, worn campaign badge/sticker, taken part in demonstrations, and member of trade union), and norms and effective sanctions (feel safe and victim of burglary/assault).

The ESS, a large-scale biennial survey of attitudes and values consisting of seven rounds/ waves that date back to 2002, provided information for all 13 aspects of social capital we investigated. The ESS data were gathered in face-to-face interviews conducted in the native language of the interviewees. The interviewees were asked questions about various dimensions of social capital aspects and had to provide answers that ranged on a scale from 0 to 10, depending on the specific question. Information on social capital is available at http://www.europeansocialsurvey.org and described in detail in the Appendix.

2.2 Testable hypotheses

In this section, we argue that social capital, in terms of social and institutional trust, networks, and norms and effective sanctions, has on average a positive dual role on immigration sentiments – a direct one, as expressed by H_1 hypothesis and an indirect one, as expressed by H_2 hypothesis, both presented in the following.

A crucial dimension of social capital is trust. Social trust is defined as trust in strangers, trust in people with whom we are not previously acquainted. Institutional trust is placing confidence in the working and mechanisms of society's institutions. Trust creates bonds between people, influences individual outcomes as well as social group interactions, facilitates cooperation and efficiency, and enhances the quality of the political, legal, and institutional environment (North, 1990), increasing thus society's overall effectiveness (Paxton, 1999; Paldam, 2001). Nevertheless, a more pessimistic possibility is that social trust could be confined just to strangers close to the truster's cultural background (Hero, 2003). We will empirically test this idea in our article, but we argue here, following theoretical justifications, that people can develop expectations of trust about other people from different cultural and ethnic backgrounds. Rather than behaving from an imperative moral rule that people are trustworthy, the social truster is willing to offer cooperation to a stranger and then evaluates his or her behavior depending on the result. This makes the social truster to be more exposed to risk than the distruster, but the truster can also achieve higher pay-offs if his or her cooperative moves are reciprocated by the stranger. As a result, trusters can obtain "social intelligence" as a consequence of their cooperative moves and learn to discern better which people are trustworthy. Consequently, they tend to avoid the use of cues and heuristics based on racist or cultural stereotypes, because his or her "social intelligence" teaches him or her that these cues and heuristics are systematically biased (Yamagishi, 2001).

Networks is another important aspect of social capital. Interpersonal social networks such as meeting colleagues, friends, or family lead to a greater social embeddedness of the individual and to a strong feeling of belonging in the society and thus enhance trust and cooperation. Arguably, societies with citizens who are more engaged in the community life are tolerant and respectful of others and their rights and show altruistic preferences and positive attitudes on behalf of others and of the wider community (Mansbridge, 1999).⁹ Voluntary associations act as sources of tolerant social views. People who differ in many ways (concerning their cultural, educational and income background), except for the special interest that their association serves, meet as equals on a voluntary basis and interact in pursuit of common concerns. Widely diversified networks, especially diversified weaker ties, increase contact with many different people and backgrounds and enhance tolerance (Putnam, 2000).

The negative flip side of the contact argument is the competition argument. Whereas cooperative contact fosters tolerance, competition and lack of personal contact foster intolerance, and this can trigger a darker side of social capital. When ethnic groups compete in certain settings, they perceive each other as threats.¹⁰ Where market competition with minority groups is high, such associations can increase awareness of competition pressure, heighten the perception of group threat, and lead to relatively low levels of tolerance (Côté and Erickson, 2009). We will empirically test this idea in our article, but we argue here that voluntary associations are vehicles of democratic and learning processes (Perrin, 2005), where people trust each other, learn a range of civic skills and virtues, and develop "social intelligence" through cooperation. Through contact and diverse linkages, members of associations develop experience, skills, and mechanisms to become less affected from racist and cultural stereotypes.

Finally, another aspect of social capital is that of solid norms and transparent and effective sanctions, which reduce the incentives for criminal action; individuals who do not feel afraid but feel safe in the surroundings they live in develop stronger ties within their community, feel less threat, and are likely to overcome the uncertainty associated with the unknown.

Overall, rich social capital portrays societies that are more equalitarian, cooperative, and integrated (Putnam, 2000) with better governance (Bowles and Gintis, 2002) and higher economic prospects (Knack and Keefer, 1997). Consequently, immigration and ethnic diversity would be less of a challenge to community cohesion in societies where social capital is present. Therefore, the first hypothesis we aim to test is the following:

H_i: Social capital tends to relate to more positive sentiments toward immigration.

The literature on public opinion on immigration has analyzed various versions of two main hypotheses: the "perceived threat" hypothesis, which argues that racial prejudice toward immigrants is a response to a perceived threat to the economic and political privileges of the

⁹ A more pessimistic possibility, however, is that social trust is confined just to strangers close to the truster's cultural background.

¹⁰ For example, occupational groups, including labor unions and business and professional associations, bring together people experiencing similar job pressures and tensions.

dominant ethnic group in a society (Blumer, 1958), and the "contact" hypothesis, which, in contrast, considers that the presence of large populations of out-groups provides members of the dominant group with first-hand experience of different people, which, under certain conditions, can diminish racial prejudices (Oliver and Wong, 2003).

Social capital may have an additional, indirect role on shaping immigration attitudes in a society; it can mitigate typical "perceived threat" conditions in a society such as country's economic conditions (Oliver and Mendelberg, 2000) or personal economic circumstances (Olzak, 1992), and relative size of the subordinate groups (Blalock, 1967). Even in situations in which cultural or racial prejudices would be more salient, people would feel threat due to their own conditions, or in times of economic upheavals when usually people feel threat of losing privileges or being heavily taxed, societies with high levels of trust in people and institutions, effective social networks, and solid norms and transparent sanctions will be more immune to the saliency of such prejudices. Based on this claim, we can test the second hypothesis:

 H_2 : Social capital tends to mitigate the negative effects of "perceived threat" on attitudes toward immigrants.

The task of the following sections is to develop the model and econometric techniques to test for these two hypotheses.

2.3 Modeling individual's attitudes toward immigration

Suppose that stated immigration concerns depend on latent continuous concerns about immigration. If y^* denotes latent concerns and y the stated concerns, then the following holds:

$$y_{it}^{\star} = x^T \beta + e_{it} \tag{1}$$

where *i* and *t* denote country and time, *x* is a vector of covariates, and β is the vector of regression coefficients to be estimated.

Furthermore, suppose that while we cannot observe y^* , we can instead observe the categories of response:

$$y = \begin{cases} 0, \text{ if } y^* \le c_1, \\ 1, \text{ if } c_1 < y^* \le c_2, \\ 2, \text{ if } c_2 < y^* \le c_3, \\ \vdots \\ N, \text{ if } c_N < y^* \end{cases}$$
(2)

To capture the complex and multifaceted nature of sentiments toward immigration, we consider the dependent variable (y^*) to be a vector of interviewees' responses in six different statements: three about the effects of immigration on economic, cultural, and welfare prospects of the host country and three about the ethnic/race background of the immigrant.

The vector *x* is a set of covariates that includes *sociodemographic*, *social capital*, *macro-economic* and *foreign participation* variables. More analytically, the set *sociodemographic* contains characteristics of the individual, namely, gender, age, education, marital status, health status, net income, domicile, employment status, religiousness, political orientation, and the origin (family roots) of the interviewee. The literature in sociology has identified some of these characteristics as important ones in shaping attitudes (Semyonov et al., 2006; Hainmueller

and Hiscox, 2007). Arguably, one would expect that older, low-income, low-educated, unemployed, and extreme right in political orientation individuals, for example, would express more xenophobic opinions about immigration. This set of variables has been commonly used in all micro-level analyses with evidence, on their statistical importance, to vary across studies.

The next set of variables is what the literature calls social capital and here we include three important dimensions: (i) trust in people and institutions, (ii) interpersonal networks, and (iii) norms and effective sanctions. As it has been argued in previous sections, one expects a positive association of all three dimensions of social capital and attitudes toward immigration. Empirical evidence corroborates to the important role of trust in shaping positive attitudes toward immigration (Herreros and Criado, 2009; Rustenbach, 2010).

The set *macroeconomic* includes variables that characterize the economic environment of a country that could also shape public assessments. Arguably, public attitudes toward immigrants become more positive in good economic times and more negative in economic down-turns.¹¹ We consider a number of variables, namely, the GDP per capita, unemployment, central government debt, social benefits, the Gini coefficient to proxy economic welfare, fiscal burden, and (in)equality conditions of the host country. Empirical evidence of the macroeconomic conditions, however, is weak and sometimes with perverse results (Schneider, 2008; Rustenbach, 2010). We further consider the share of elderly people in a country to proxy country's demographics. As rising old-age dependency ratios pose a challenge to the viability of public pension and health systems, the loosening of immigration policies is often seen as one policy option to counter this challenge. The share of elderly people in a country tests whether sentiments toward immigration differ across countries with different age distributions. Empirical evidence (Calahorrano, 2013) supports that over the life cycle, the stated immigration concerns are predicted to increase well into retirement and decrease afterward.

The set *foreign participation* contains a number of variables relevant to the multicultural and ethnic diversity of the host country. The presence of foreign stock in a country and country's colonial tradition may influence immigration attitudes as well. According to a simplified version of contact theory, increased contact with immigrants should undermine xenophobic sentiment; people who live next to well-integrated neighbors tend to have positive immigration sentiments (Ha, 2010). In contrast, threat theory argues that intergroup contact intensifies conflict due to competition over scarce resources; from this perspective, in countries with larger immigrant populations, there is more of a perceived group threat, leading to more antipathy toward the out-group (Hjerm, 2009). Empirical evidence, however, is mixed as to the relationship between the presence of foreigners and individuals' perceptions toward immigration.¹² Colonial tradition influences migration patterns between former imperial powers and their colonies of the past.¹³ It intends to capture whether sentiments to immigration differ

¹¹ At economic downturns, the fiscal cost of immigration becomes more an issue of concern and raises worries among individuals, who are either at the bottom of the income distribution and thus are threatened of immigrant competition in the labour market and welfare cuts or further up of the income distribution and therefore worry about the potential tax implications of immigration-induced expansion of the welfare budget.

¹² For example, Schlueter and Wagner (2008) found that between regions, a larger size of the immigrant population increases negative reactions, but within regions, more immigrants increase intergroup contact and reduce immigrant derogation. In a similar vein, Markaki and Longhi (2013) found that regions with a higher percentage of immigrants born outside the EU and a higher unemployment rate among the immigrant population show higher probability for negative attitudes to immigration. In contrast, Rustenbach (2010) found no evidence.

¹³ Clear examples are the cases of old colonial countries such as the UK and France and immigrant nations like the US where many earlier immigrants have now become citizens or are second- or third-generation "immigrants".

between countries with colonial history (and potentially richer and more accustomed to ethnic diversities) and countries with no such history at all (Gallaher et al., 2009).

2.4 Estimation strategy

Our dependent variable, attitude of the individuals toward immigration, is ordinal; that is, we can rank the values, but the real distance between categories is unknown. Respondents state that immigration is "bad for a country's economy" if their latent concern exceeds a higher threshold *c*1, "good for a country's economy" if their latent concern exceeds a much higher threshold *c*2, and "very good for a country's economy" if their latent concern exceeds an even higher threshold *c*3.¹⁴ The vector parameter β and *c*=(*c*1, *c*2, *c*3)' can then be chosen such as to maximize the likelihood of observing the sample on hand. This requires an assumption on the distribution of *e*. Testing the distribution of the error term, we employ ordered logistic regression model, as described in the section earlier, in which the estimated set of regression coefficients (β) predicts the probability of the outcome of interest. We estimate our ordered logit using maximum likelihood estimation (MLE) techniques.

A critical concern in any empirical analysis is endogeneity, which produces biased and inconsistent estimates. Omitted factors such as trade collaborations, networks (Eichengreen and Irwin, 1998), and historic linkages (Anjonami and Hariri, 1992) between countries of origin and destination not captured in our estimations may induce endogeneity issues. To alleviate such concerns, we use country and year fixed effects. Furthermore, aspects of social capital, especially trust in people can be endogenous and bias our estimates. To account for endogeneity issues, we apply a control function approach (Wooldridge, 2015) that has the advantage of primarily dealing with nonlinear models as it is our case. In doing so, we regress the instruments on our endogenous variable in the first stage, conditional upon the other exogenous variables of the original model, and recover the predicted residuals of this estimation to plug them, at the second stage, into our original model (without excluding our endogenous variables from the model). Our inference is based on bootstrapping over all two-step procedure and for 1,000 iterations. Statistically insignificant predicted residuals would be an indication that there is no statistically significant bias at the coefficients of social capital due to endogeneity.

We use the variable "feel fair treated" as an instrument for our endogenous variable "Trust People", which is an important dimension of social capital. The choice of "feel fair treated" is justified based on theory and statistics. On the theory front, both fairness and trust are crucial to our willingness to cooperate with others. Smith and Lindsay (2014) argued that leaders aiming to build trust in relationships need to pay attention to the issue of fairness. The direct interplay between the two was studied by Müller et al. (2008) who explored the causal link between procedural fairness and trust. Their results demonstrated that changes in perceived fairness of a procedure lead to changes in the willingness to entrust. This finding illustrates that trust judgments and trustful behavior can be influenced by procedural fairness judgments. On statistical front, a good instrument needs to be strongly correlated with the endogenous variable under instrumentation (i.e., "Trust People" in our case) and little or no related at all

¹⁴ Immigration is considered "very bad for a country's economy" if individuals' latent concern is below threshold cl.

with the error term and the depended variable (i.e., immigration attitude or immigration policy). In other words, "feel fair treated" is a reasonable choice for an instrument if it can only be correlated with the dependent variable, immigration attitudes, through its effect on the endogenous variable "Trust People". Indeed, the correlation between the instrument ("feel fair treated") and the endogenous variable "Trust People" is 0.57 and is statistically significant at 5%, while the correlation between the instrument and the estimated error is 0.03 and is statistically significant at 5%.

3 Data description and analysis

Our empirical analysis covers 22 countries for the period 2002 to 2014 and a total of 267,282 individuals. Table A1, in the Appendix, presents the participating countries and the number of observations per country in each wave of survey. Analytical descriptions of all variables and data sources are presented in the Section "Data" of the Appendix.

As summary statistics reveal in Table A2 in the Appendix, Europeans have, on average, a rather positive stance on immigration and different immigrant profiles, as the scores of the six dependent aspects of the dependent variable indicate. In some respect, this contrasts with the mixed attitudes, dualities in thinking, and splits on immigration issues expressed, for instance, in studies for the US (Suro, 2009). Europeans tend to welcome more immigrants of the same/ race and less from poorer non-European countries. Cultural cohesion seems to be more of a concern than economic or life quality impacts of immigration. On average, Europeans are more social trusters than distrusters in others and their institutions, socialize often, are not much of church-goers, and the majority have not worked in political or action groups, contacted politicians and government officers, displayed campaign budge, participated in public demonstrations, or have been member of trade unions; they feel rather safe and have not often fallen victims of burglary/assault over the last 5 years. Finally, on average, 0.1% of the population of a European country applied for asylum, 0.19% have acquired the nationality of the country, about 6.6% are foreigners (foreign born, immigrants of first and second generation), and almost half of the countries have been colonizers some period in their history with at least one colony.

To obtain more insights, Figure 1 maps Europe over our sample period according to the responses of the Europeans on the statement "immigration is bad or good for country's economy" (*Economy*). Dark blue indicates more xenophile sentiments and very light blue less.

As one can see, Switzerland, Sweden, and Norway show the most xenophile attitude toward immigration, while Greece, Hungary, and Czech Republic the least. This finding is consistent for all different immigration statements. The composition of individuals' sentiments per country and over sample period is portrayed in Figure A1 of the Appendix.

Next, we explore the trend of Europeans' attitudes over time. Table 1 shows the evolution of the attitudes toward "immigration is bad or good for country's economy" across countries and time: wave 2002–2004 (beginning of the survey), wave 2006–2008 (beginning of financial crisis), and wave 2012–2014 (Syrian exodus).

Despite the challenges Europe has been through, Europeans have a positive stance toward immigration with about half of the countries to rank above sample mean. Furthermore, immigration attitudes of large countries, such as Germany, the UK, and France, that



Figure 1 European attitudes toward immigration is bad or good for country's economy.

Table 1	Country trends toward	l immigration	("immigration	is bad/good	for a	country's
	economy")					

Country	Wave 2002–2004	Wave 2006-2008	Wave 2012–2014
Austria	2.89	-	2.61
Belgium	2.47	2.55	2.42
Czech Republic	2.44	2.31	2.13
Denmark	2.54	2.72	2.56
Estonia	-	2.47	2.64
Finland	2.73	2.80	2.71
France	2.72	2.57	2.45
Germany	2.70	2.70	2.89
Greece	2.14	2.04	-
Hungary	2.33	2.08	2.12
Ireland	2.63	2.60	2.64
Italy	2.79	-	-
Netherlands	2.61	2.76	2.56
Norway	2.79	2.84	2.84
Poland	2.49	2.83	2.61
Portugal	2.55	2.66	2.67
Slovak Republic	-	2.41	-
Slovenia	2.41	2.39	2.31
Spain	2.79	2.73	2.65
Sweden	2.84	2.84	2.89
Switzerland	2.97	3.06	3.05
United Kingdom	2.44	2.53	2.59
Sample mean	2.61	2.59	2.60

Note: Respondents choose among points: 1=very bad, 2=bad, 3=good, and 4=very good.

have been traditionally immigrant-receiving nations and suffered from terrorist attacks show diverging stances: UK and France score below sample mean, with France becoming more xenophobic, while Germany has remained xenophile with an even increasing trend. With respect to the Southern countries that were hit the hardest by the financial crisis, the attitudes also differ significantly: Spain shows a rather positive but declining stance over immigration, Portugal appears to be more welcoming over time, and there is a strong declining tendency for Greece. For the latter, there is a sharp pick around the year 2004 – as Greece hosted the Olympic Games that year – and a sharp decline thereafter, as the economic conditions deteriorated ("Greek crisis"). Unfortunately, Greece lacks data for the 2012–2014 wave and Italy for the 2006–2008 and 2012–2014 waves. Figure A2, in the Appendix, visualizes some of these findings.

Finally, we explore how immigration sentiments alter in response to economic conditions, presence of foreigners in a country, and level of social capital. The left graph of Figure 2 associates standards of living – proxied by GDP per capita (y-axis) and immigration attitudes (x-axis), while the right graph associates social capital – proxied by trust in people (y-axis) and immigration sentiments. Bubbles' size in both graphs equals to the size of foreigners to population in the host country.

A number of interesting patterns emerge: First, the higher the standards of living, the less xenophobic a country is. Switzerland has the largest GDP per capita and the largest share of foreigners (size of the bubble) and shows the least opposition to immigration, while Greece, Hungary, Slovenia, and Czech Republic have the lowest GDP per capita and the highest anti-immigration sentiments, with Greece (and Switzerland) being more multicultural (about 6% of their population are foreigners, period average) compared to the aforementioned countries. Second, high social truster societies are more xenophile. For example, Swedish show about two times more trustworthiness to each other and to their institutions than the Greeks, who, together with the Hungarians, Slovenians, and Czech Republic people, show the lowest score in trust.

4 Empirical results

This section presents our results. First, we examine what factors shape Europeans' sentiments toward immigrants, and second, whether these sentiments vary across different immigrant profiles.

Figure 2	Sentiments toward immigration, standards of living, social trust, and size of for-
	eign stock (%).



2a: Sentiments towards Immigration, Standards of Living and Size of Foreign-stock (%)" 2b: Sentiments towards Immigration, Social Trust and Size of Foreign-stock (%)"

4.1 What explains Europeans' xenophile(/phobic) attitudes toward immigrants?

Table 2 shows the results. Columns (i) to (iii) report odd estimates of equation (1) on the interviewees' attitudes toward "immigration is bad or good for country's economy" (*economy*), "country's cultural life undermined or enriched by immigrants" (*culture*), and "immigrants make country worse or better place to live" (*place to live*), respectively. Heteroscedasticity-adjusted standard errors are reported in parentheses.

The estimates above are odd ratios. One can read the odd ratios as follows: if the odd ratio, a, is bigger than one (a > 1), then the probability of an individual to express very positive sentiments toward immigration ($y_{it} = 4$, i.e., maximum level) increases by (a - 1)*100%, whereas the probability decreases by (1 - a)*100%, if the odd ratio is smaller than one (a < 1). The coefficient of the control term included tells us furthermore that the latent factor captured by the instruments is correlated with our dependent variable(s). Note, however, that the bias is small and the control term is not statistically significant, so endogeneity does not impose a serious concern.

In terms of individual characteristics, the educational level of the individual matters the most. Higher levels of formal education tend to be associated with positive attitude toward immigration's effects on economy (70.3%), culture (73.5%), and making a country a better place to live (65.3%).¹⁵ The origin of the respondent's parents could make the individual more tolerant to cultural differences and immigration (Goldstein and Peters, 2014). Furthermore, living in a city/town, high income level, good health status, and religiousness of the individual are also positively associated with a friendlier stance toward immigration.¹⁶ In contrast, unemployed people and blue collar workers may feel a general threat from new comers, due to their situation, which is at "risk", and are more likely to express anti-immigration sentiments (Herreros and Criado, 2009; Markaki and Longhi, 2013). Compared to people in paid jobs, jobless individuals are about 17% more probable to have a negative stance on immigration as they feel threat from increased job competition (Mayda, 2006; Gorodzeisky, 2011). This finding is consistent with rational competition theories; employment status has always been a crucial predictor of attitudes to minorities. Political ideology, as expected, also associates with stances toward immigration. In addition, gender (being a woman) also appears to be associated with anti-immigration attitudes. Marital status (being married versus all other alternatives) and age play little role in shaping attitude as the coefficients are statistically insignificant in almost all specifications. However, the older the individuals get, the more xenophobic they become as the relevant estimates of age in columns (ii) and (iii) indicate.

The next set of estimates analyze the role of social capital. Among its three dimensions, it is the trust in people and institutions (legal, health, and education) that has the largest weight in shaping positive sentiments toward immigrants. It has been argued that social trusters have altruistic preferences (Mansbridge, 1999). Social trust implies an expectation that strangers

¹⁵ Higher educated individuals are less likely to express prejudice, negative stereotypes, and racism toward minorities (Herreros and Criado, 2009) for at least two reasons. First, according to the labour market competition theory, as immigrants mostly work in low-skilled manual jobs, they are more likely to be complement rather than substitute to highly educated natives (Bogard and Sherrod, 2008). Second, the link between education and attitudes is deeply rooted in the fact that educational systems tend to promote acceptance of different cultural values and beliefs (Hainmueller and Hiscox, 2007).

¹⁶ High levels of income provide financial security, good health status makes one feel psychologically better and not being in competition for social security benefits, and religiousness cultivates positive feelings towards the mankind and promotes treating people equally.

Variable	Economy	Culture	Place to live
Gender	0.832*** (0.022)	1.066 (0.045)	1.028 (0.038)
Age	1.004 (0.018)	0.934*** (0.012)	0.925*** (0.014)
Education level	1.703*** (0.069)	1.735*** (0.071)	1.653*** (0.062)
Marital status	0.987 (0.077)	0.969 (0.028)	0.933* (0.037)
Health status	1.083*** (0.025)	1.088*** (0.022)	1.107*** (0.016)
Income level	1.158*** (0.040)	1.075*** (0.029)	1.070*** (0.023)
Domicile	1.187*** (0.041)	1.202*** (0.063)	1.204*** (0.043)
Employment status	0.833** (0.071)	0.921* (0.044)	0.802*** (0.058)
Religiousness	1.052*** (0.017)	1.077*** (0.029)	1.076*** (0.024)
Political orientation	0.761*** (0.035)	0.676*** (0.039)	0.697*** (0.037)
Immigrant parents	1.649*** (0.139)	1.560*** (0.076)	1.742*** (0.121)
Trust people	2.792*** (0.127)	3.302*** (0.329)	3.157*** (0.334)
Trust legal system	1.159*** (0.059)	1.140 (0.117)	1.066 (0.117)
State of health services	1.314*** (0.054)	1.065 (0.067)	1.219*** (0.059)
State of education system	1.044 (0.030)	1.044 (0.039)	1.090*** (0.030)
Social interaction	1.010 (0.017)	1.054*** (0.019)	1.001 (0.016)
Church attendance	1.176*** (0.055)	1.099 (0.077)	1.264*** (0.085)
Worked in political/ action group	1.419*** (0.091)	1.279** (0.127)	1.326*** (0.088)
Contacted politician/ government	1.096** (0.044)	1.082** (0.034)	1.075* (0.046)
Worn campaign badge/sticker	1.138*** (0.024)	1.254*** (0.039)	1.199*** (0.041)
Taken part in demonstrations	1.524*** (0.068)	1.619*** (0.068)	1.480*** (0.056)
Member of trade union	0.931 (0.050)	1.024 (0.039)	0.971 (0.035)
Feel safe	1.409*** (0.058)	1.398*** (0.072)	1.529*** (0.064)
Victim of burglary/assault	1.025 (0.019)	1.082** (0.040)	1.034 (0.030)
D _{HighGDPcap}	1.350*** (0.264)	1.427 (0.409)	1.303 (0.374)
D _{HighDebt}	0.553* (0.171)	0.753 (0.192)	0.639 (0.208)
D _{HighElderlyPop}	0.790 (0.132)	1.121 (0.390)	1.100 (0.381)
D _{HighAsylumSeekers}	0.425*** (0.094)	0.598 (0.239)	0.508* (0.193)
D _{HighForeignStock}	0.491*** (0.088)	0.899 (0.322)	0.777 (0.260)
D _{HighNationalityAcq}	0.779 (0.184)	0.832 (0.278)	1.055 (0.310)
D _{ColonialTradition}	1.220 (0.260)	0.844 (0.208)	0.671 (0.227)
$D_{HighGDPcap\&HighForeignStock}$	2.975*** (0.729)	0.820 (0.325)	0.942 (0.367)
D _{HighTrust&HighAsylumSeekers}	1.159* (0.101)	1.057 (0.166)	1.223 (0.177)
D _{HighTrust&HighForeignStock}	1.206*** (0.069)	1.084 (0.132)	1.327*** (0.146)
D _{HighTrust&HighNationalityAcq}	1.055 (0.104)	1.234 (0.206)	1.098 (0.148)
D _{HighTrust&Unemployment}	1.029 (0.057)	1.132** (0.058)	1.175** (0.086)
D _{HighTrust&LowIncome}	1.103** (0.053)	0.998 (0.042)	1.062** (0.029)
Control term	0.753 (0.278)	0.745 (0.238)	0.932 (0.356)
Observations	165,173	165,772	165,331

Table 2 Estimates of Europeans' attitude toward immigrants (odd ratios)

Notes: Estimates are odd ratios. Control function approach is used to correct for endogeneity in "Trust People" (see the Estimation Section). Control term is the predicted residual of the control function estimation. Country and year fixed effects are used. Heteroscedasticity-adjusted standard errors are given in parentheses. *, **, and *** are significance levels at 10%, 5%, and 1%, respectively.

are trustworthy, and this expectation is extended to all people, including immigrants from different cultural backgrounds. The "mechanism" that links social trust and positive attitudes toward immigrants is that trusters exclude heuristics and cues based on racial or cultural stereotypes when forming their beliefs about other people's trustworthiness. As our results show, the coefficient related to trust in people is significant and in the expected direction. Social trust is associated with more positive attitude toward immigrants and their impact on economy (179.2%), culture (230.2%), and quality of living (215.7%) than social distrust. This result is also confirmed by the study of Herreros and Criado (2009). Likewise, institutional trust, especially trust in the legal system, shapes positive attitudes to immigration. Trust in the legal system associates with 15.9% higher probability of positive attitude to immigrants' effect on economy (14% and 6.6%, respectively, on culture and quality of living). Satisfaction from the health and education systems also associates with 31.4% and 4.4%, respectively, more positive sentiments to immigrants' effect on economy. Institutional trust increases society's overall effectiveness (Paxton, 1999; Paldam, 2001), makes people more confident for the society they live in and the way the society copes with stressful situations. If immigration is such, then trusters in country's institutions, all other variables considered, tend to express more welcoming attitude toward immigrants compared to institutional distrusters.

Social networks of informal type (meeting with friends, relatives, and colleagues; participating in associations including churches; and voluntary work) or formal type (work for political parties or professional organizations among others) are also found to positively associate with immigration attitude index. Individuals who are socially and, especially, politically active tend to express more positive immigration sentiments compared to individuals who are not. Among these activities, being member or working for political parties/action groups, participating in campaigns and lawful public demonstrations, significantly increases the probability of having a more positive attitude toward immigrants. The only exception is the association with trade union. In times of widespread unemployment, competition between indigenous and immigrant workers might increase, making inclusive union policies difficult to maintain (Penninx and Roosblad, 2000). However, trade union estimates are statistically insignificant. Solid norms and transparent and effective sanctions reduce the incentives for criminal action; individuals who feel safe and not being victimized develop stronger ties within their community and feel less threat. Consequently, absence of criminality is highly associated with positive stance toward immigrants. This conjecture is confirmed by our estimates of feeling safe that carry the expected sign and are statistically significant.

Overall, we find a strong, positive direct association of almost all aspects of social capital on immigration sentiments of the European citizens. This finding validates our first hypothesis (*H1*) that social capital associates with positive attitudes toward immigration across European countries.

We continue with macroeconomic conditions. We find that countries with higher GDP per capita ($D_{HighGDPcap}$) are 35% more probable to associate with positive attitude toward immigrants' effect on economy than countries with lower GDP per capita. The fiscal cost of immigration also appears to be an important aspect of attitude. Countries with high debt ($D_{HighDebl}$) are about 45% more prone to associate with anti-immigration sentiments compared to countries with low debt. For example, immigrants who do not find employment are a fiscal burden on the welfare state and thus for natives of all ages. In contrast to the literature that provides

weak or even mixed evidence on the link of macroenvironment on public sentiments, we find strong and clear results.¹⁷ Furthermore, we find that countries with a high share of elderly population tend to exhibit negative sentiments over immigrants' economic impact, but they keep a positive stance on immigrants' influence on local culture and quality of living.

The stock of foreigners contributes to the multicultural profile of a country. The latter may stir up both positive and negative reactions: it can encourage the decrease of prejudice or can increase the perception of threat to national identity. Our results show that citizens in countries with many foreigners ($D_{HighForeignStock}$) and refugees/asylum seekers ($D_{HighAsylumSeekers}$) are about 50.9% and 57.5%, respectively, more prone to show anti-immigrant feelings compared to citizens who live in countries with more ethnically homogenous population and less refugees (Markus, 2014). For some people, perhaps, there can be a threshold beyond which they do not think favorably toward new immigrants. This negative association between the size of immigrant (foreign) population and immigration feelings is also documented in the literature (Oliver and Mendelberg, 2000). The number of nationality acquisitions ($D_{HighNationalityAcq}$) and the colonial tradition ($D_{ColonialTradition}$) of the country are not statistically significant.

Nevertheless, the negative association between multiculturalism and immigration attitude may alter once we control for the level of economic development of the country. The coefficient of the dummy $D_{HighGDPcap&HighForeignStock}$ captures exactly this. What we observe is that countries with a high percentage of foreign population and high GDP per capita sentiments toward immigrants are more positive, especially when it comes to the economic impact, than countries with a high percentage of foreign population and low GDP per capita; a finding consistent with Figure 2. Wealthier countries have better economic and institutional mechanisms to provide for and integrate different ethnicities/races. Independent of economic conditions, however, individuals' prejudice may remain against foreigners, and this is indeed the case as the coefficient of $D_{HighGDPcap&HighForeignStock}$ turns negative in the other two specifications, losing, however, its statistical significance.

Having analyzed all factors (and their interactions) that could possibly shape immigration sentiments, we can now embark on testing our second hypothesis (*H2*) that social capital smoothes out the negative effects of adverse social conditions on attitudes toward immigration. That is, irrespective of a country's social conditions, when we consider a country's social capital (here, level of social trust), the picture that emerges may contradict with the xenophobic stance documented above.

According to our findings, in countries that host big numbers of foreigners or asylum seekers and are rich in social trust, the sentiments toward immigrants tend to be more xenophile than in countries with low social trust. The coefficient of $D_{HighTrust\&HighForeignStock}$ captures this positive association – larger than 1 in all specifications and statistically significant in the last one. This is because countries rich in social capital are more cooperative and egalitarian (Putnam, 2000; Bowles and Gintis, 2002) and, therefore, able to cope effectively with accommodating new comers. Social trusters are also 15.9% ($D_{HighTrust\&HighAsylumSeekers}$) more prone to have a positive stance on the impact of asylum seekers on the quality of living. Similar evidence emerges when we revisit the individual characteristics and particularly the ones that relate to

¹⁷ Empirical evidence on the macroeconomic conditions, however, is mixed, mainly weak and sometimes with perverse results. For instance, Semyonov et al. (2006) found that higher GDP per capita implies significantly more negative attitudes, while Schneider (2008) and Hatton (2016) supported the opposite relation. Other studies (Rustenbach, 2010) found no impact at all.

the "threat effect". This time, however, we consider how different is a social truster individual who is unemployed ($D_{HighTrust\&Unemployment}$) or a social truster individual who has a very low income level ($D_{HighTrust\&LowIncome}$) from equal peers who are social distrusters. The results alter dramatically. Unemployed trusters are not negative at all to immigrants compared to unemployed distrusters. In fact, they are 13.2% and 17.5% more probable to have positive attitude on the immigrants' impact on culture and quality of living, respectively. Likewise, low-income social trusters are about 10.3% more positive to immigrants than low-income distrusters. Such findings corroborate with the claim of our second hypothesis on the positive indirect association between social capital and individuals' attitudes toward immigration.

Overall, our empirical analysis documents the important role of almost all aspects of social capital on public sentiments about immigration; the evidence on social trust corroborates findings of related studies of the past (Herreros and Criado, 2009; Rustenbach, 2010). The two testable hypotheses find strong support. Social capital has a positive association with immigration attitudes via two ways, a direct way, social capital associates with positive attitude toward immigration, and an indirect way, social capital moderates the negative effects of "perceived threat" on people's opinions about immigrants. Individuals with low income, unemployed people, or citizens in multicultural and ethnically diverse environments, where the risk of intergroup conflicts is high, will nonetheless exhibit generally positive attitudes toward immigration if they are social trusters.

A potential issue in our analysis is that the average positive correlation between associations and immigration sentiments could mask important variation across different types of associations. Erickson (2004) and Côté and Erickson (2009) argued that associations that are heterogeneous tend to relate to more social tolerance than homogenous ones. They proposed and constructed a measure of network and association diversity based on income class, occupation, and gender of their members. As we do not have information on the composition of associations and on particular types of associations (e.g., ethnic associations for minority ethnic groups) for the European citizens, we can only derive nuanced conclusions.¹⁸

4.2 Do Europeans' sentiments vary across different immigrant profiles?

In this section, we turn our attention into examining whether sentiments toward immigrants differ depending on immigrant's profile, that is, race/ethnicity and country of origin. Different attitude toward different types of immigrants reveals the concerns of the Europeans toward certain cultures, and it is an indication of individuals' perception toward the immigration policy (stricter or looser toward certain immigrant profiles) their country should apply.

Table 3 shows the results. Columns (i) to (iii) report the odd ratios of equation (1) on the respondent's attitudes toward "allow many/few immigrants of same race/ethnic group as majority" (*same race*), "allow many/few immigrants of different race/ethnic group from majority" (*dif-ferent race*), and "allow many/few immigrants from poorer non-European countries" (*poorer/non-EU*), respectively. Heteroscedasticity-adjusted standard errors are reported in parentheses.

All sociodemographic variables that were important in the previous analysis pertain their statistical significance carrying the expected sign; however, in some cases, their coefficients

¹⁸ The literature argues that the specific consequences of social capital largely depend on political and social conditions, the existence of inequality and poverty, and liberal democracies, to name a few (Gargiulo and Benassi, 1999; van Deth and Zmerli, 2010; Villalonga-Olives and Kawachi, 2017). In our analysis, we included most of these factors to control the effect of social capital on immigration attitudes.

Variable	Same race	Different race	Poorer/non-EU
Gender	0.974 (0.016)	1.004 (0.028)	1.051** (0.025)
Age	0.946** (0.026)	0.895*** (0.021)	0.876*** (0.016)
Education level	1.641*** (0.061)	1.728*** (0.061)	1.562*** (0.043)
Marital status	0.836*** (0.058)	0.854** (0.055)	0.939 (0.064)
Health status	1.116*** (0.019)	1.110*** (0.016)	1.075*** (0.018)
Income level	1.156*** (0.020)	1.115*** (0.022)	1.096*** (0.027)
Domicile	1.145*** (0.042)	1.176*** (0.038)	1.107*** (0.036)
Employment status	0.843** (0.067)	0.802* (0.092)	0.930 (0.126)
Religiousness	1.046 (0.036)	1.021 (0.034)	1.066* (0.037)
Political orientation	0.752*** (0.027)	0.669*** (0.032)	0.666*** (0.031)
Immigrant parents	1.352*** (0.060)	1.335*** (0.075)	1.292*** (0.065)
Trust people	2.445*** (0.244)	2.295*** (0.197)	2.052*** (0.146)
Trust legal system	0.905 (0.101)	0.937 (0.118)	0.908 (0.106)
State of health services	1.080 (0.061)	1.072 (0.058)	1.061 (0.045)
State of education system	0.964 (0.035)	0.959 (0.033)	0.942 (0.036)
Social interaction	1.009 (0.023)	0.995 (0.029)	1.025 (0.032)
Church attendance	1.187*** (0.067)	1.269*** (0.087)	1.343*** (0.095)
Worked in political/action group	1.232*** (0.053)	1.272*** (0.096)	1.289*** (0.077)
Contacted politician/government	1.139*** (0.042)	1.127*** (0.052)	1.074* (0.040)
Worn campaign badge/sticker	1.223*** (0.024)	1.266*** (0.048)	1.247*** (0.037)
Taken part in demonstrations	1.361*** (0.045)	1.509*** (0.082)	1.508*** (0.079)
Member of trade union	0.966 (0.035)	0.981 (0.037)	0.982 (0.040)
Feel safe	1.244*** (0.050)	1.344*** (0.042)	1.297*** (0.029)
Victim of burglary/assault	1.054 (0.036)	1.049 (0.038)	1.072** (0.032)
$D_{HighGDPcap}$	1.359 (0.444)	1.255 (0.455)	1.292 (0.511)
D _{HighDebt}	0.788 (0.222)	0.580 (0.240)	0.631 (0.293)
D _{HighElderlyPop}	1.170 (0.340)	1.173 (0.504)	1.056 (0.501)
$D_{HighAsylumSeekers}$	0.771 (0.316)	0.471* (0.208)	0.450 (0.221)
D _{HighForeignStock}	1.026 (0.295)	0.671 (0.241)	0.673 (0.270)
$D_{HighNationalityAcq}$	0.598 (0.195)	0.977 (0.368)	0.908 (0.384)
D _{ColonialTradition}	1.015 (0.468)	1.560 (0.760)	1.493 (0.788)
$D_{HighGDPcap\&HighForeignStock}$	1.174 (0.518)	2.073 (1.024)	2.323 (1.297)
$D_{HighTrust\&HighAsylumSeekers}$	1.221 (0.149)	1.205 (0.160)	1.192 (0.162)
D _{HighTrust&HighForeignStock}	1.324** (0.172)	1.285* (0.172)	1.228 (0.169)
$D_{HighTrust\&HighNationalityAcq}$	1.017 (0.134)	1.100 (0.151)	1.172 (0.173)
D _{HighTrust&Unemployment}	1.081 (0.057)	1.167 (0.110)	0.991 (0.066)
D _{HighTrust&LowIncome}	1.075* (0.040)	1.059** (0.029)	1.063** (0.030)
Control term	0.676 (0.231)	0.912 (0.382)	0.965 (0.465)
Observations	165,779	165,772	165,573

 Table 3
 Estimates of Europeans' attitude toward different immigrant types (odd ratios)

Notes: Estimates are odd ratios. Control function approach is used to correct for endogeneity in "Trust People" (see the Estimation Section). Control term is the predicted residual of the control function estimation. Country and year fixed effects are used. Heteroscedasticity-adjusted standard errors are given in parentheses. *, **, and *** are significance levels at 10%, 5%, and 1%, respectively.

change across different immigrant profiles as they become more negative to immigrants from different race/ethnicity and especially to those from poorer non-European countries; note-worthy exception is the highly educated and religious individuals who actually become more receptive to very different racial/ethnic background.

Furthermore, all three dimensions of social capital continue to greatly contribute to the xenophile attitude of the individuals, a finding that supports our first hypothesis. Our results show that when social capital is present, there is no differentiation at all across different origins of immigrants.

Macroeconomic conditions do not seem to shape Europeans' sentiments favorably toward any group of immigrants, in particular as estimates tend to be statistically insignificant. Perhaps, macroeconomic conditions do not directly relate to feelings about the ethnic background of the immigrants but rather to their skills. Unfortunately, we have no information about the skills of the immigrants relative to those of natives. This information could shed some light as documented in the study of Mayda (2006). When it comes to the statistical significance of the interaction terms of social capital with the macroeconomic variables, the same picture emerges. Perhaps, this is because we have no deeper information about the social capital that would allow us to unfold different reactions based on the ethnic background of the immigrants. For instance, information on associations of ethnic minorities, their contact with the natives, and their embeddedness into the society would shed more light why native citizens are more (or less) friendly to different immigrant backgrounds. This requires detailed information that is not available at the moment. Future research could focus further on these issues, disentangling potential diversity in the effects of social capital.

When it comes to multiculturalism, it seems that it does not pose any threat when social trust is present as the coefficient of $D_{HighTrust\&HighForeignPop}$ indicates. In such societies, immigrants of all types are welcome. Furthermore, unemployed trusters do not actually differentiate among immigrants. Same holds for low-income social trusters. Both findings are in line with the claim of our second hypothesis and therefore validate it.

Finally, endogeneity is not a serious concern here as the control term is statistically insignificant.

In sum, a consistent finding that emerges is that public attitude toward immigrants appears to vary across different immigrant profiles. Europeans, on average, are slightly more xenophile compared to immigrants of the same race/ethnicity. Social capital tends to associate with pro-immigration attitudes and diminishes any differentiation across immigrant profiles.

4.3 Robustness

For robustness purposes, we make an effort to instrument for the social capital as a whole and not just for the societal trust. In doing so, we construct, using factor analysis, an index of social capital based on all of its individual components. Then, we instrument for it. This is not an easy task, though. To find appropriate and exogenous instruments, we resort to the economic growth literature. Potential candidates would be historic data (the past affects the present, but not the other way around) about a country's literacy rates and dominant religion in the ninetieth century (i.e., as early as we have data from the establishment of the modern nations). Historic educational endowments represent a good predictor of current institutional differences across countries of Europe. As Glaeser et al. (2004) had shown, historic levels of schooling are robustly correlated with changes in political institutions and economic development. Furthermore, Tabellini (2010) had demonstrated that literacy in 1880 represents an important source of variation for the cultural traits and level of social capital of European countries. Religion has also been a proxy for culture and widely used in the finance and economics literature (La Porta et al., 1999) as it is a key component of a system of beliefs. For instance, the literature argues that the way a religion is organized, i.e., hierarchical vs. equalitarian structures, reflects on how the society and government are structured (centralization vs. individualism; Stulz and Williamson, 2003).

We therefore adopt the same instrumental variables, i.e., countries' literacy rates (percentage of population) and dominant religion (e.g., Catholic, Protestants, Greek Orthodox, Muslim) in the 1900s as an exogenous determinant of current social capital quality. Data are retrieved from the following sources: religion is obtained from the World Factbook (2019) and literacy rates from Roser and Ortiz-Ospina (2020). In addition, we also include, along with these two instruments, the "feel fair treated" variable to allow for individual (and not only country) variation in our instruments. We then apply the control function approach (Wooldridge, 2015). That is to say, we regress the instruments on our dependent variable in the first stage, conditional upon our instruments, and recover the predicted residuals of this estimation to plug them into our original model (without excluding the constructed index of social capital); inference is based on bootstrapping over all two-step procedure, 1,000 iterations. The results of this process are shown in Table A3 of the Appendix.

The coefficient of the control term tells us furthermore that the latent factor captured by the instruments is correlated with our dependent variable(s). Hence, endogeneity seems to cause a small bias in the social capital coefficients in our previous estimates. Note, however, that the bias is small and the control term is not significant, so the main conclusions of the analysis undertaken, so far, hold; that is, social capital positively relates to immigration sentiments and further mitigates the negative effects of perceived threat on people's opinions about immigrants.

We also performed a battery of sensitivity analysis to sharpen the robustness of our results. We split our sample in two sub-periods, 2002–2008 and 2010–2014, to test whether financial crisis had an impact on public sentiments about immigration. The results did not alter significantly.

We also included various control variables, namely, the Gini coefficient for inequality and welfare benefits (% of GDP). The estimates are statistically insignificant, and the rest of the results are barely modified.

Overall, the results do not change in any significant way across different specifications, subsamples, and alternative definitions.

5 Conclusion

Large international migration flows over the past decades have fundamentally changed the landscape of the European countries' populations and brought immigration to the front of the research and policy agenda. The question of how public opinion toward immigration is formed has become as relevant as ever with implications for both present and potential immigrants as well as for policymakers to meet the challenges associated with increased immigration.

This article has studied the European attitudes toward immigration and the contextual factors that shape these attitudes. Based on recent and detailed data on 267,282 respondents over

the period 2002–2014, we find that despite the eventful past years, Europeans, on average, are still positive toward immigrants with the North European countries being the least xenophobic. A salient finding of our analysis, however, is regardless of the impact of other contextual factors, individuals with a high social capital do exhibit more positive attitudes toward all immigrants, independently of their background, than the rest of the population. Furthermore, social capital moderates the negative effects of "perceived threat" on people's opinions about immigration.

The implication of our results is straightforward. A successful public policy should aim to increase social cohesion and invest in trust across groups and institutions.

Declarations

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Competing interests

None

Funding

This work was supported by the European Commission Framework - Horizon 2020 Grant (H2020-EURO-2014-2015/649378-FIRES). The role of the funding body was mainly in the design and analysis of the study.

Author contributions

Claire Economidou composed the manuscript, Dimitrios Karamanis analyzed and interpreted the data, Alexandra Kehrinioti collected the data, Sofia Xesfingi reviewed the relevant literature. The Authors declare that they have read and approved the final manuscript.

Acknowledgments

We are grateful to George Bouloucheris for excellent research assistance. We also thank Alexandros Bechlioulis, Kyriakos Drivas, Sotiris Karkalakos and Dimitris Konstantios and several seminar and conference participants for useful insights.

References

Alesina, A.; E. La Ferrera (2000): Who Trusts Others? Journal of Public Economics 85(2), 207-234.

Anjonami, A.; V. Hariri (1992): Migration Stock and the Issue of Competing and Complementary Flows in United States Interstate Migration. Journal of Population Economics 5(2), 87-100.

Arrow, K.J. (1972): Gifts and exchanges. Philosophy & Public Affairs 1(4), 343-362.

- Arrow, K.J. (1999): Observations on Social Capital, in: Dasgupta, P.; I. Serageldin (eds.), Social Capital: A Multifaceted Perspective, Washington: World Bank.
- Bellamare, C.; S. Kroger (2004): On Representative Social Capital. European Economic Review 51, 183-202. Blalock, H. (1967): Toward a Theory of Minority-Group Relations. New York: John Wiley & Sons.

Blumer, H. (1958): Race Prejudice as a Sense of Group Position. Pacific Sociological Review 1(Spring), 3-7.

- Bogard, K.; I. Sherrod (2008): Citizenship Attitudes and Allegiances in Diverse Youth. Cultural Diversity & Ethnic Minority Psychology 14(4), 286-296.
- Bourdieu, P. (1986) The forms of capital. In J. Richardson (Ed.) Handbook of Theory and Research for the Sociology of Education (New York, Greenwood), 241-258.
- Bowles, S.; H. Gintis (2002): Social Capital and Community Governance. Economic Journal 112(127), 419-436.
- Brader, T., Valentino, N., Suhay, E. (2008): What Triggers Public Opposition to Immigration? Anxiety, Group Cues, and Immigration Threat. American Journal of Political Science 52(4), 959-978.
- Bruni, L.; R. Sugden (2000): Moral Canals: Trust and Social Capital in the Work of Hume, Smith and Genovesi. Economics & Philosophy 16, 21-45.
- Calahorrano, L. (2013): Population Aging and Individual Attitudes Towards Immigration: Disentangling Age, Cohort and Time Effect. Review of International Economics 21, 342-353.
- Card, D.; C. Dustmann; I. Preston (2012): Immigration, Wages, and Compositional Amenities. Journal of the European Economic Association 10(1), 78-119.

Coleman, J. (1988): Social Capital in the Creation of Human Capital. American Journal of Sociology 94, S95-S120. **Coleman, J.S.** (1990): Foundations of Social Theory. Cambridge: Harvard University Press.

- Côté, R.R.; B.H. Erickson (2009): Untangling the Roots of Tolerance: How Forms of Social Capital Shape Attitudes Toward Ethnic Minorities and Immigrants. American Behavioral Scientist 52(12), 1664-1689.
- Dinesen, P.; R. Klemmensen; A. Nergaard (2016): Attitudes Toward Immigration: The Role of Personal Predispositions. Political Psychology 37(1), 55-72.
- Durlauf, S.N. (1999): The Case 'Against' Social Capital. Focus 20(3), 1-5.

- Durlauf, S.N.; M. Fafchamps (2004): Social Capital. NBER Working Paper 10485. Massachusetts: National Bureau of Economic Research.
- Dustmann, C.; K. Vasiljeva; A. Pill Damm (2019): Refugee Migration and Electoral Outcomes. The Review of Economic Studies 86(5), 2035-2091.
- Eichengreen, B.; D.A. Irwin (1998): The Regionalisation of the World Economy, in: Frankel, J.A. (ed.), The Role of History in Bilateral Trade Flows. Chicago: University of Chicago Press, 33-57.
- Erickson, B.H. (2004): The Distribution of Gendered Social Capital in Canada, in: Flap, H.; B. Volker (eds.), Creations and Returns of Social Capital: A New Research Program. New York: Routledge.
- Facchini, G.; A.M. Mayda (2009): Does the Welfare State Affect Individual Attitudes Toward Immigrants? Evidence Across Countries. The review of Economics and Statistics 9(12), 295-314.
- Fine, B. (2002): The F**k you up those Social Capitalists. Antipode 34(4), 796-799.

Fine, B. (2010): Theories of Social Capital: Researchers Behaving Badly. London: Pluto Press.

- Fitzgerald, J.; K. Curtis; C. Corliss (2012): Anxious Publics Worries About Crime and Immigration. Comparative Political Studies 45(4), 477-506.
- Ford, R. (2011): Acceptable and Unacceptable Immigrants: How Opposition to Immigration in Britain is Affected by Migrants' Region of Origin. Journal of Ethnic and Migration Studies 37(7), 1017-1037.
- Gallaher, C.; C.T. Dahlman; M. Gilmartin; A. Mountz; P. Shirlow (2009): Colonialism/imperialism. Key Concepts in Political Geography. London: SAGE Publications.
- Gannon, B.; J. Roberts (2018): Social Capital: Exploring the Theory and Empirical Divide. Empirical Economics 1–21. https://doi.org/10.1007/s00181-018-1556-y.
- Gargiulo, M.; M. Benassi (1999): The Dark Side of Social Capital, in: Leenders, R.T.A.J.; S.M. Gabbay (eds.), Corporate Social Capital and Liability. Boston: Kluwer, 298-322.
- Glaeser, E.L.; R. La Porta; F. Lopez-de-Silanes; A. Shleifer (2004): Do Institutions Cause Growth? Journal of Economic Growth 9, 271-303.
- Goldstein, J.; M. Peters (2014): Nativism or Economic Threat: Attitudes Toward Immigrants During the Great Recession. International Interactions 40, 376-401.
- Gorodzeisky, A. (2011): Who Are the Europeans That Europeans Prefer? Economic Conditions and Exclusionary Views toward European Immigrants. International Journal of Comparative Sociology 52(1-2), 100-113.
- Ha, S.E. (2010): The Consequences of Multiracial Contexts on Public Attitudes Toward Immigration. Political Research Quarterly 63(1), 29-42.
- Hainmueller, J.; D. Hangartner (2015): Does Direct Democracy Hurt Immigrant Minorities? Evidence from Naturalization Decisions in Switzerland. Available at SSRN: https://ssrn.com/abstract=2022064 or http:// dx.doi.org/10.2139/ssrn.2022064.
- Hainmueller, J.; M. Hiscox (2007): Educated Preferences: Explaining Individual Attitudes Toward Immigration in Europe. International Organization 61, 399-442.
- Hatton, T. (2016): Immigration, Public Opinion and the Recession in Europe. Economic Public Policy 31(86), 205-246. Hero, R.E. (2003): Social Capital and Racial Inequality in America. Perspectives in Politics, 1(1), 113-22.
- Herreros, F.; H. Criado (2009): Social Trust, Social Capital and Perceptions of Immigration. Political Studies 57, 337-355.
- Hjerm, M. (2009): Anti-Immigrant Attitudes and Cross-Municipal Variation in the Proportion of Immigrants. Acta Sociologica 52(1), 47-62.
- Hopkins, D. (2011): The Upside of Accents: Language Differences and Attitudes Toward Immigration. British Journal of Political Science 45(3), 531-557.
- Kan, K. (2007): Residential Mobility and Social Capital. Journal of Urban Economics 61, 436-457.
- Knack, S.; P. Keefer (1997): Does Social Capital Have an Economic Payoff? A Cross-Country Investigation. The Quarterly Journal of Economics 112(4), 1251-1288.
- Kreps, D.M.; P. Milgrom; J. Roberts; R. Wilson (1982): Rational Cooperation in the Finitely Repeated Prisoners' Dilemma. Journal of Economic Theory 27, 245-252.
- La Porta, R.; F. Lopez-de-Silanes; A. Shleifer; R.W. Vishny (1999): The Quality of Government. Journal of Law, Economics, and Organization 15, 222-279.
- Lahav, G.; M. Courtemanche. (2012): The Ideological Effects of Framing Threat on Immigration and Civil Liberties. Political Behavior 34(3), 477-505.
- Mansbridge, J. (1999): Altruistic Trust, in: Warren, M. (ed.), Democracy and Trust. New York: Cambridge University Press, 290-309.
- Markaki, Y.; S. Longhi (2013): What Determines Attitudes to Immigration in European Countries? An Analysis at the Regional Level. Migration Studies 1, 311-337.

Markus, A. (2014): Attitudes to Immigration and Cultural Diversity in Australia. Journal of Sociology 50(1), 10-20.

Mayda, A.M. (2006): Who is Against Immigration? A Cross-Country Investigation of Individual Attitudes Toward Immigrants. The Review of Economics and Statistics 88(3), 510-530.

- McDaniel, E.; I. Nooruddin; A. Shortle (2011): Divine Boundaries: How Religion Shapes Citizens' Attitudes Toward Immigrants. American Politics Research 39(1), 205-233.
- Müller, P.A.; Greifeneder, R.; Stahlberg, D.; Van den Bos K.; Bless, H., 2008. The Role of Procedural Fairness in Trust and Trustful Behavior (No. 08-21). Sonderforschungsbereich 504, Universität Mannheim & Sonderforschungsbereich 504, University of Mannheim.
- North, D. (1990): Institutions, Institutional Change and Economic Performance. New York: Cambridge University Press.
- Oliver, J.; T. Mendelberg (2000): Reconstructing the Environmental Determinants of White Racial Attitudes. American Journal of Political Science 44(3), 574-589.
- Oliver, J.E.; J. Wong (2003): Intergroup Prejudice in Multiethnic Settings. American Journal of Political Science 47(4), 567-582.

Olzak, S. (1992): The Dynamics of Ethnic Competition and Conflict. Standford CA: Standford University Press. Paldam, M. (2001): Corruption and Religion Adding to the Economic Model. Kyklos 54(2-3), 383-413.

- Paxton, P. (1999): Is Social Capital Declining in the United States? A Multiple Indicator Assessment. American Journal of Sociology 105, 88-127.
- Penninx, R., Roosblad, J. (2000): Trade Unions, Immigration, and Immigrants in Europe, 1960-1993: A Comparative Study of the Attitudes and Actions of Trade Unions in Seven West European Countries. New York/ Oxford: Berghahn Books.

Perrin, A.J. (2005): Political Microcultures: Linking Civic Life and Democratic Discourse. Social Forces 84(2), 1049-1082. Portes, A.; P. Landolt (1996): The Downside of Social Capital. The American Prospect 26, 18-22.

- Putnam, R.D. (1993): What Makes Democracy Work? National Civic Review 82(2), 90-209.
- Putnam, R.D. (2000): Bowling Alone. The Collapse and Revival of American Community. New York: Simon and Schuster.
- Quillian, L. (1995): Prejudice as a Response to Perceived Group Threat: Population Composition and Anti-Immigrant and Racial Prejudice in Europe. American Sociological Review 60(4), 586-611.
- Rocco, R.; E. Fumagelli (2014): The Empirics of Social Capital and Health, in: Folland, S.; L. Rocco (ed.), The Economics of Social Capital and Health: A Conceptual and Empirical Roadmap. World Scientific Series in Global Health Economics and Public Policy. Singapore: World Scientific Publishing, 93-152.
- Roser, M.; E. Ortiz-Ospina (2020): "Literacy". Published online at OurWorldInData.org (https://ourworldindata. org/literacy).
- Rustenbach, E. (2010): Sources of Negative Attitudes Toward Immigrants in Europe: A Multilevel Analysis. International Migration Review 44, 53-77.
- Sabatini, F. (2008): Social Capital and the Quality of Economic Development. Kyklos 61(3), 466-499.
- Sabatini, F. (2009): Social Capital as Social Networks: A New Framework for Measurement and an Empirical Analysis of its Determinants and Consequences. Journal of Socio-Economics 38, 429-442.
- Schlueter, E., Wagner, U. (2008): Regional Differences Matter: Examining the Dual Influence of the Regional Size of the Immigrant Population on Derogation of Immigrants in Europe. International Journal of Comparative Sociology 49(2-3), 153-173.
- Schneider, S. (2008): Anti-Immigrant Attitudes in Europe: Outgroup Size and Perceived Ethnic Threat. European Sociological Review 24, 53-67.
- Semyonov, M.; R. Raijman; A. Gorodzeisky (2006): The Rise of Anti-Foreigner Sentiment in European Societies. American Sociological Review 71, 426-449.

Smith, J.G.; Lindsay J.B., 2014. Fairness and trust. In Beyond inclusion (pp. 117-146). Palgrave Macmillan, New York.

Stulz, R.M.; R. Williamson (2003): Culture, Openness, and Finance. Journal of Financial Economics 70, 313-349.

- Suro, R. (2009): America's Views of Immigration: The Evidence from Public Opinion Surveys. Migration Policy Institute (MPI), Report, January.
- Tabellini, G. (2010): Culture and Institutions: Economic Development in the Regions of Europe. Journal of the European Economic Association 8, 677-716.
- The World Factbook (2019): Washington, DC: Central Intelligence Agency, 2019. https://www.cia.gov/library/ publications/resources/the-world-factbook/index.html.
- Van Deth, J.W.; S. Zmerli (2010): Introduction: Civicness, Equality, and Democracy A "Dark Side" of Social Capital? The American Behavioral Scientist 53(5), 631-639.
- Villalonga-Olives, E.; I. Kawachi (2017): The Dark Side of Social Capital: A Systematic Review of the Negative Health Effects of Social Capital. Socical Science & Medicine 194, 105-127.
- Wooldridge, J.M. (2015): Control Function Methods in Applied Econometrics. Journal of Human Resources 50(2), 420-445.
- Yamagishi, T. (2001): Trust as A Form of Social Intelligence, in: Cook, K. (ed.), Trust in Society. New York: Russell Sage Foundation, 121-47.

Appendix

A.1 Data

The main source of our data is the European Social Survey (ESS), a large-scale biennial study of attitudes and values, consisting of seven rounds/waves that date back to 2002. The ESS data were gathered in face-to-face interviews conducted in the native language of the interviewees. All rounds together contained micro data for almost 336,964 individuals; 267,282 individuals were employed in our analysis. Data are available at http://www.europeansocialsurvey.org. Regarding data availability, as Table A1 shows, some countries in the data set are represented with observations from all ESS waves, while others only from some waves. Raw data were adjusted using poststratification and population size weights, provided by the ESS to control for qualitative characteristics of the interviewees in each wave within a country and for different country sizes. A detailed discussion on weights is available at https://www.europeansocial-survey.org/docs/methodology/ESS_weighting_data_1.pdf.

Country				ESS wa	ves			
	Country code	2002 (obs)	2004 (obs)	2006 (obs)	2008 (obs)	2010 (obs)	2012 (obs)	2014 (obs)
Austria	AUT	2,257	2,256	2,405	-	-	_	1,795
Belgium	BEL	1,899	1,778	1,798	1,760	1,704	1,869	1,769
Switzerland	CHE	2,040	2,141	1,804	1,819	1,506	1,493	1,532
Czech Republic	CZE	1,360	3,026	-	2,018	2,386	2,009	2,148
Germany	DEU	2,919	2,870	2,916	2,751	3,031	2,958	3,045
Denmark	DNK	1,506	1,487	1,505	1,610	1,576	1,650	1,502
Estonia	EST	-	1,989	1,517	1,661	1,793	2,380	2,051
Spain	ESP	1,729	1,663	1,876	2,576	1,885	1,889	1,925
Finland	FIN	2,000	2,022	1,896	2,195	1,878	2,197	2,087
France	FRA	1,503	1,806	1,986	2,073	1,728	1,968	1,917
United Kingdom	GBR	2,052	1,897	2,394	2,352	2,422	2,286	2,264
Greece	GRC	2,566	2,406	-	2,072	2,715	-	-
Hungary	HUN	1,685	1,498	1,518	1,544	1,561	2,014	1,698
Ireland	IRL	2,046	2,286	1,800	1,764	2,576	2,628	2,390
Italy	ITA	1,207	-	-	-	-	960	-
Netherlands	NLD	2,364	1,881	1,889	1,778	1,829	1,845	1,919
Norway	NOR	2,036	1,760	1,750	1,549	1,548	1,624	1,436
Poland	POL	2,110	1,716	1,721	1,619	1,751	1,898	1,615
Portugal	PRT	1,511	2,052	2,222	2,367	2,150	2,151	1,265
Sweden	SWE	1,999	1,948	1,927	1,830	1,497	1,847	1,791
Slovenia	SVN	1,519	1,442	1,476	1,286	1,403	1,257	1,224
Slovakia	SVK	-	1,512	1,766	1,810	1,856	1,847	-
Total		38,308	41,436	36,166	38,434	38,795	38,770	35,373

Table A1 Number of obs by country and wave

Note: Obs: observations; Min: minimum; Max: maximum; ESS: European Social Survey; EU: European Union; GDP: gross domestic product; PPP: purchasing power parity; WDI: World Development Indicators; CIA: Central Intelligence Agency; OECD: Organisation for Economic Cooperation and Development.

The dependent variable (y) is a vector and contains individuals' responses on six statements about immigration: (i) "immigration is bad or good for country's economy", (economy), (ii) "country's cultural life undermined or enriched by immigrants" (culture), (iii) "immigrants make country worse or better place to live" (place to live), (iv) "allow many/few immigrants of same race/ethnic group as majority" (same race), (v) "allow many/few immigrants of different race/ethnic group from majority" (different race), and (vi) "allow many/few immigrants from poorer countries outside Europe" (poorer/non-EU). The data source for the individuals' responses is the ESS. The answers are ordinal and limited. For the first three statements, the respondents were shown a card with a 10-point scale, where only the two most distant points were articulated (i.e., 0 - bad for the economy and 10 - good for the economy, 0 - cultural life undermined and 10 - cultural life enriched, and 0 - worse place to live and 10 - better place to live, respectively), while for the last three statements, the interviewees had to choose among four points (1 - allow many, 2 - allow some, 3 - allow a few, and 4 - allow none). We reduced the 10-point scale of the first three variables (economy, culture, and place to live) to a 4-point scale based on the quantile distribution of the answers provided. In this way, we have less classes and all six dependent variables are expressed to a 4-point scale.

A number of regressors are included in the vector set *x*. Information on the sociodemographic variables (*sociodemographic*) – gender (*gender*), age (*age*), education level (*education level*), marital status (*marital status*), health status (*health status*), net income level (*income level*), domicile (*domicile*), employment status (*employment status*), religiousness (*religiousness*), political orientation (*political orientation*), and whether one of the parents was immigrant (*immigrant parents*) – is also derived from the ESS database. In almost all aforementioned variables, we "merged" the range of a respondent's categories just to reduce the number of classes.

Data on different dimensions of social capital (*social capital*) – trust in people (*trust people*) and in institutions (*trust legal system*, *state of health services*, *state of education system*), networks (*social interaction, church attendance, worked in political/action group, contacted politician/government, worn campaign badge/sticker, taken part in demonstrations*, and *member of trade union*), and norms and effective sanctions (*feel safe* and *victim of burglary/assault*) – are retrieved from the ESS as well. In a similar vein, as with the sociodemographic variables, we modified the ESS range of interviewees' responses.

Data on macroeconomic variables (*macroeconomic*), GDP per capita (constant 2011 thousand dollars, purchasing power parity) (*GDPcap*), and central government debt ratio to GDP (*debt/GDP*) come from the World Bank, *World Development Indicators* database. Elderly people share to country's population comes from the Central Intelligence Agency (World Factbook, 2019) database and is a dummy that takes the value of 1 if a country's elderly population share is above sample's median, and 0 otherwise.

The set of foreign participation (*foreign participation*) includes variables relevant to the multicultural profile of a country, namely, percentage of foreigners (foreign born, first and second generation) to country's population, percentage of asylum seekers to country's population, and percentage of nationality acquisitions to country's population – all derived from the Organisation for Economic Cooperation and Development. Information on the colonial tradition of a country is extracted from Gallaher et al. (2009).

In order to cut the data in various ways, all macroeconomic and foreign participation variables enter in our model as dummies. For example, $D_{HighGDPcap}$ takes the value of 1 if a country's GDP per capita is above sample's median, and 0 otherwise, and $D_{HighDebt}$ and $D_{HighElderbyPop}$ get the .

value of 1 if a country's debt ratio and elderly's population ratio, respectively, are above sample's median, and 0 otherwise. Similarly, the variables in the foreign participation set become also dummies, $D_{HighForeignStock}$, $D_{HighAsylumSeekers}$, and $D_{HighNationalityAcq}$, and take the value of 1 if the percentage of foreigners to country's population, percentage of asylum seekers to country's population, percentage of nationality acquisitions to country's population, respectively, is above sample's median, and 0 otherwise. The dummy $D_{ColonialTradition}$ takes the value of 1 if a country has been a colonizer, even once in its history, and 0 otherwise.

Variable	Source	Obs	Mean	Standard	Min	Мах
				deviation		
<i>y</i> variables						
Economy	ESS	256,331	2.591	0.935	1	4
Culture	ESS	256,850	2.561	1.071	1	4
Place to live	ESS	256,441	2.614	0.892	1	4
Same race	ESS	259,601	2.778	0.849	1	4
Different race	ESS	259,392	2.507	0.872	1	4
Poorer/non-EU		258,859	2.437	0.890	1	4
<i>x</i> variables						
Sociodemographic						
Gender	ESS	267,005	0.532	0.499	0	1
Age	ESS	267,282	3.754	1.694	1	6
Education level	ESS	265,758	2.279	0.466	1	3
Marital status	ESS	267,282	0.298	0.457	0	1
Health status	ESS	266,995	2.584	0.635	1	3
Income level	ESS	199,430	1.845	0.736	1	3
Domicile	ESS	266,503	1.625	0.484	1	2
Employment Status	ESS	267,282	0.055	0.227	0	1
Religiousness	ESS	265,197	1.945	0.819	1	3
Political orientation	ESS	235,269	1.924	0.750	1	3
Immigrant parents	ESS	266,854	0.144	0.352	0	1
Social capital						
Trust people	ESS	266,438	0.644	0.479	0	1
Trust legal system	ESS	260,985	0.640	0.480	0	1
State of health services	ESS	263,854	0.649	0.477	0	1
State of education system	ESS	254,428	0.552	0.497	0	1
Social interaction	ESS	266,668	0.623	0.485	0	1
Church attendance	ESS	266,088	0.256	0.436	0	1
Worked in political/ action group	ESS	266,492	0.040	0.196	0	1
Contacted politician/ government	ESS	266,424	0.147	0.354	0	1

Table A2Summary statistics

(Continued)

Variable	Source	Obs	Mean	Standard deviation	Min	Мах
Worn campaign badge/ sticker	ESS	266,294	0.077	0.267	0	1
Taken part in demonstrations	ESS	266,388	0.066	0.248	0	1
Member of trade union	ESS	263,952	0.424	0.494	0	1
Feel safe	ESS	264,863	0.776	0.417	0	1
Victim of burglary/ assault	ESS	266,551	0.188	0.390	0	1
Feel fair treated as instrument	ESS	265,389	0.537	0.499	0	1
Macroeconomic						
GDP per capita (constant 2011, PPP)	WDI	267,282	38,801	18,877	8,815	90,807
Debt (% GDP)	WDI	267,282	57.67	28.51	4.400	146.2
Elderly population (% population)	CIA	267,282	18.47	2.071	12.84	21.76
Foreign participation						
Asylum seekers (% population)	OECD	267,282	0.101	0.120	0.0005	0.774
Stock of foreigners (% population)	OECD	267,282	6.623	4.784	0.0164	23.78
Nationality acquisition (% population)	OECD	267,282	0.189	0.148	0.0026	0.624
Colonial tradition	Gallaher et al. (2009)	267,282	0.39	0.488	0	1

Table A2	(Continued)
----------	-------------

Note: Obs: observations; Min: minimum; Max: maximum; ESS: European Social Survey; EU: European Union; GDP: gross domestic product; WDI: World Development Indicators; OECD: Organisation for Economic Cooperation and Development.



Figure A1 Country attitudes toward immigration is bad or good for country's economy.



Figure A2 Trends of European attitudes toward immigration.

	Economy	Culture	Place to live	Same race	Different race	Poorer/non-EU
Gender	0.827*** (0.0288)	1.091 (0.0612)	1.020 (0.0507)	0.962 (0.0382)	0.975 (0.0418)	1.014 (0.0461)
Age	1.006 (0.0208)	0.939*** (0.0118)	0.932*** (0.0141)	0.950* (0.0266)	0.900*** (0.0208)	0.879*** (0.0166)
Education Level	1.771*** (0.0752)	1.774*** (0.0707)	1.712*** (0.0718)	1.770*** (0.0962)	1.876*** (0.0962)	1.713*** (0.0826)
Marital status	0.987 (0.0712)	0.986 (0.0267)	0.942 (0.0355)	0.834*** (0.0549)	0.854** (0.0534)	0.932 (0.0613)
Health status	1.112*** (0.0262)	1.081*** (0.0227)	1.131^{***} (0.0352)	1.152*** (0.0436)	1.149*** (0.0437)	1.123*** (0.0406)
Income level	1.168*** (0.0354)	1.081*** (0.0238)	1.070*** (0.0164)	1.163*** (0.0214)	1.121*** (0.0236)	1.102^{***} (0.0298)
Domicile	1.163*** (0.0453)	1.203*** (0.0736)	1.172*** (0.0463)	1.129*** (0.0379)	1.148*** (0.0377)	1.083** (0.0364)
Employment status	0.845** (0.063)	0.821** (0.062)	0.888 (0.0859)	0.838** (0.0680)	0.800** (0.0893)	0.920 (0.124)
Religiousness	1.107*** (0.0270)	1.104^{**} (0.0434)	1.148^{***} (0.0338)	1.478 (0.492)	1.478 (0.495)	1.267 (0.416)
Political orientation	0.757*** (0.0367)	0.664*** (0.0398)	0.693*** (0.0384)	0.742*** (0.0281)	0.658*** (0.0331)	0.655*** (0.0327)
Immigrant parents	1.611^{***} (0.149)	1.492*** (0.0840)	1.695^{***} (0.130)	1.314*** (0.0577)	1.304^{***} (0.0814)	1.264*** (0.0749)
Social capital	2.101*** (0.398)	2.611*** (0.547)	2.352*** (0.666)	1.101^{***} (0.0406)	1.090^{**} (0.0383)	1.154^{***} (0.0373)
DHinhGDPran	1.142*** (0.0321)	1.319 (0.424)	1.246 (0.395)	1.354 (0.462)	1.254 (0.478)	1.307 (0.536)
DHinhDeht	1.125*** (0.0407)	0.654 (0.225)	0.731 (0.266)	0.784 (0.249)	0.586 (0.263)	0.619 (0.309)
DHiahElderlyPon	0.759 (0.139)	1.060 (0.387)	1.021 (0.375)	1.144 (0.347)	1.138 (0.510)	1.032 (0.509)
D High Acvlum Seekers	0.407*** (0.109)	0.561 (0.233)	0.492* (0.208)	0.768 (0.344)	0.466 (0.225)	0.447 (0.233)
D High Foreign Stock	0.451*** (0.0936)	0.773 (0.277)	0.705 (0.242)	1.056 (0.329)	0.682 (0.263)	0.703 (0.300)
DHiahNationalityAca	0.741 (0.214)	0.738 (0.280)	0.956 (0.325)	0.611 (0.196)	0.982 (0.371)	0.944 (0.398)
DeclenialTradition	1.273 (0.338)	0.894 (0.249)	0.697 (0.285)	1.037 (0.514)	1.574~(0.830)	1.497 (0.829)
D HighGDPcan&HighEoreign Stock	3.312*** (0.907)	1.004 (0.406)	1.118(0.480)	1.324 (0.576)	2.362* (1.137)	2.616* (1.402)
D High Social Captures High Avelum Seekers	1.085** (0.0354)	1.067 (0.180)	1.175 (0.175)	1.121 (0.112)	1.104 (0.126)	1.082 (0.127)
D High Social Capitals High Foreign Stock	1.199*** (0.0664)	1.073 (0.115)	1.254*** (0.0877)	1.166* (0.0955)	1.134(0.0934)	1.070 (0.0961)
D High Social Capitals, High National ity Aca	1.031 (0.113)	1.201 (0.209)	1.064 (0.143)	0.957 (0.121)	1.037 (0.138)	1.101(0.154)
D High Social Contral & Une molovment	1.142*** (0.0426)	1.139*** (0.0427)	1.133^{***} (0.0416)	1.043 (0.0590)	1.120 (0.105)	0.950 (0.0588)
Designment of the second s	1.099** (0.0433)	0.997 (0.0451)	1.092*** (0.0278)	1.111*** (0.0272)	1.102^{**} (0.0432)	1.137*** (0.0328)
Control term	0.826 (0.139)	0.619** (0.124)	0.738 (0.182)	0.965 (0.284)	0.998 (0.291)	1.114(0.321)
Observations	165,173	165,772	165,331	165,779	165,772	165,573
<i>Notes</i> : Estimates are odd ratios. of the control function estimatio	Control function approa	ach is used to correct fo <i>capital</i> constructed ind	or endogeneity in <i>social</i> lex: "literacy in 1900s", '	<i>capital</i> constructed inc "dominant religion", ar	dex. Control term is th nd "feel fair treated". C	e predicted residuals ountry and year fixed

 Table A3
 Social capital (as a factor of 13 variables), control function estimates

Robustness

effects are used. Heteroscedasticity-adjusted standard errors are given in parentheses. *, **, and *** are significance levels at 10%, 5%, and 1%, respectively.